Author’s response to reviews

Title: Associations between social capital and maternal depression: results from a follow-up study in China

Authors:
Chi Zhou (zhc.kathy@163.com)
Weijun Zheng (deardangjun@163.com)
Qi Yuan (Qi_YUAN@imh.com.sg)
Baodan Zhang (2512203004@qq.com)
Hao Chen (824743@qq.com)
Weijue Wang (439136687@qq.com)
Liu Huang (364200374@qq.com)
Liangwen Xu (lwxu2006@163.com)
Lei Yang (yanglei62@hznu.edu.cn)

Version: 1 Date: 23 Mar 2017

Author’s response to reviews:

Dear Editor and Reviewers:

Thank you for your letter and for the reviewer’s comments concerning our manuscript entitled “Social Capital and Postpartum Depression: a 6-Month Study in Zhejiang, China” (ID: PRCH-D-17-00218). Those comments are all valuable and very helpful for revising and improving our paper. We have studied comments carefully and have made correction which we hope meet with approval. The main corrections in the paper and the responds to the reviewer’s comments are as flowing:

1. Response to comment: It contains some plagiarised material. In particular, material from your first three references are included (unchanged) in your own text. Those authors may have, themselves, plagiarised from these two papers:
Response: We have get and read these two papers, and add them to our references. We remove the first three references, and expunge all potentially plagiarised material in our paper. All the references have been carefully checked in our paper.

2. Response to comment: Please revise the declaration section in your manuscript as necessary.

Response: The declaration section has been checked, and our data are available on the additional file 1 submitted together with the manuscript.

We tried our best to improve the manuscript, and highly appreciate for Editors/Reviewers’ warm work earnestly. Thanks again for your comments and suggestions.